Author's response to reviews

Title: Pelvic digit as a rare cause for chronic hip pain and functional impairment: a case report

Authors:

Marc Maegele (Marc.Maegele@t-online.de)

Version: 2 Date: 16 August 2008

Author's response to reviews:

First of all I deeply apologize for the significant delay in responding to the request for minor revisions of the manuscript. To be honest, I somewhat lost track of the manuscript as I thought it already went to the review process.

As requested, the following changes have been made to the manuscript:

1. Information on the patient’s ethnic origin has been provided in the case presentation section of the manuscript
2. Case report has been rephrased to Case Presentation
3. Information on the patient’s ethnic origin has been provided in the case presentation section

I hope that the manuscript is now suitable for peer review;

with best regards from Germany

Marc Maegele